Identifying Predictors of Self-Perceptions of Aging Based on a Range of Cognitive, Physical, and Mental Health Indicators: Twenty-Year Longitudinal Findings from the ILSE Study

Serena Sabatini^{1*}, Jelena Sophie Siebert², Manfred Diehl³, Allyson Brothers³, Hans-Werner Wahl²

¹College of Medicine and Health, University of Exeter, Exeter, United Kingdom

²Institute of Psychology, Heidelberg University, Heidelberg, Germany

³Department of Human Development and Family Studies, Colorado State University, Fort Collins, Colorado

*Corresponding author

Address: Centre for Research in Ageing and Cognitive Health (REACH), University of Exeter, South Cloisters, St Luke's Campus, Exeter, EX12LU, UK

Telephone: 01392 726754

Email: S.Sabatini2@exeter.ac.uk

Word count of main text: 8465

References: 119

Number of data elements: 5 tables; 5 tables in supplement online material, 9 figures.

Funding

This work was supported by the University of Exeter College of Life and Environmental Sciences (School of Psychology), University of Exeter College of Medicine and Health, and the National Health and Medical Research Council Centre for Research Excellence in Cognitive Health [#1100579 to Kaarin Anstey].

Financial support for the ILSE study came from German Federal Ministry for Family, Senior Citizens, Women, and Youth Grants [314-1722-102/16] and [301-1720-295/2.] ILSE's fourth wave of data collection was supported by the Dietmar Hopp Foundation. Manfred Diehl's contribution to this article was supported by a grant from the National Institute on Aging, National Institutes of Health (R01 AG051723).

Data availability

This study was conducted using secondary data from the ILSE study

(<u>https://www.maelstrom-research.org/study/ilse</u>). ILSE data are available to investigators outside the ILSE team after request and approval.

Acknowledgments

We are grateful to the University of Exeter for funding a PhD scholarship for Serena Sabatini to carry out this work.

Authors' contributions

Serena Sabatini served as principal investigator of the research, designed the study,

conducted data analyses, and took the lead in writing the manuscript.

Hans-Werner Wahl contributed to the design of the ILSE study, design of the current study,

provided feedback and assistance with editing of the manuscript.

Jelena Sophie Siebert contributed to data collection, created the dataset for the current study and provided feedback on the draft of the manuscript.

The remaining co-authors provided feedback and assistance with editing of the manuscript.

Conflict of interest declaration

None

Preregistration

The current study design, hypotheses, and analytic plan were preregistered and circulated via email when requesting access to the ILSE dataset.

© 2021, American Psychological Association. This paper is not the copy of record and may not exactly replicate the final, authoritative version of the article. Please do not copy or cite without authors' permission. The final article will be available, upon publication, via its DOI: 10.1037/pag0000668

Abstract

Research exploring whether health predicts self-perceptions of aging has mostly focused on single predictors and has been hampered by short observational intervals. We examined whether 20-year changes in cognitive functioning, physical and mental health predict self-perceptions of aging.

We used data of 103 German participants who remained of a sample of 500 participants born in 1930-32 enrolled in the Interdisciplinary Longitudinal Study of Adult Development (ILSE) in 1993/1996 (mean age at fourth measurement wave= 82.5 years). Health indicators included six cognitive tests, objective and subjective physical health, and self-reported depression. We used a new and multidimensional (awareness of age-related gains and losses) and a wellestablished (attitudes toward own aging) measure of self-perceptions of aging. Linear regression analyses showed that, among the cognitive tests, decline in information processing speed (Digit Symbol) predicted less awareness of age-related gains and more awareness of age-related losses but not attitudes toward own aging. Decline in subjective but not objective physical health, predicted more awareness of age-related losses and negative attitudes toward own aging, but not awareness of age-related gains. Increase in depressive symptoms predicted more awareness of age-related losses and negative attitudes toward own aging, but not awareness of age-related gains. The size of associations suggests that objective cognitive decline has limited influence on older adults' self-perceptions of aging and, if so, only when the decline is related to mental slowing. Similarly, perceived physical and mental health, but not objective health, have a small-to-moderate influence on awareness of age-related losses and attitudes toward own aging.

Keywords: Awareness of age-related change, AARC-gains, AARC-losses, attitude toward own aging, self-rated health, objective health, objective cognition, depression

Public Significance Statement

The present study suggests that, compared to those older individuals who describe their own aging more positively, those who perceive their own aging more negatively have been experiencing greater mental slowing and increased depressive symptoms. These individuals also perceive a greater decline in their physical health. Asking older people about their experience of getting older may help to identify those who would benefit the most from interventions promoting health and acceptance of aging. Self-perceptions of aging (SPA) are important for successful aging. A prime example of a concept falling under the overarching term of SPA is the established attitude toward own aging scale (ATOA; Kotter-Grühn & Hess, 2012; Lawton, 1975), which captures affective and cognitive components of attitudes related to a person's own aging process. Compared to individuals with more negative SPA, those with more positive SPA are more engaged in health-related behaviors (Dutt et al., 2016b; Kornadt et al., 2019; Montepare, 2019; Sun & Smith, 2017) and, consequently, benefit from better cognitive functioning (i.e., performance on cognitive tests) and cognitive (i.e., brain health), physical, and mental health (Klusmann et al., 2017; Kornadt et al., 2019; Sabatini et al.; Sabatini et al., 2021d; Seidler & Wolff, 2017; Westerhof et al., 2014).

A better understanding of the factors that contribute to the formation of SPA across longer time intervals may help with identification of new ways of promoting healthy aging and more positive SPA in later life. Indeed, although empirical evidence extensively documented the effects of SPA on health outcomes (Brothers et al., 2019; Brothers et al., 2020), less is known about the extent to which changes in cognitive functioning and in health contribute to the formation of SPA (Diehl et al., 2021). To a large extent, relatively global and single indicators for cognitive functioning and cognitive, physical, and mental health have been used in previous research and observational intervals were relatively short. Moreover, most studies only considered unidimensional and/or global indicators of SPA. The lack of consideration of multiple indicators of health, as well as the lack of multidimensional measures of SPA at the outcome level, has hitherto prevented investigators from digging deeper into the identification of relevant predictors and their differential impact on SPA. Thus, this study focuses exactly on these issues by relying on long-term data of the Interdisciplinary Longitudinal Study of Adult Development (ILSE) spanning approximately 2 decades of observation. Furthermore, this study takes advantage of multiple indicators in each behavioral domain and considers two established indicators of SPA in parallel as outcomes. One of these outcomes was unidimensional whereas the other one was multidimensional.

Previous Research on Predictors of Self-Perceptions of Aging

First, the experience of cognitive decline is a major facet of the common understanding of aging (Heckhausen et al., 1989). Therefore, changes in cognitive functioning and the subsequent perception of cognitive decline may serve as important reminders of aging and may, therefore, influence a person's subjective interpretation of their own aging. Indeed, poorer cognitive functioning often goes along with more negative SPA (O'Brien et al., 2020; Sabatini et al., 2021b; Seidler & Wolff, 2017; Zhu & Neupert, 2020). However, perceived cognitive decline is not necessarily due to objective changes in cognitive functioning. Indeed, evidence on the association of perceived cognitive decline with objective cognitive functioning and cognitive health is inconsistent. On the one hand, several studies found that individuals reporting cognitive complaints are more likely to have experienced a decline in their cognitive functioning (e.g., poorer score on cognitive tests) and cognitive health (e.g., increased amyloid deposition in the brain) compared to those not reporting subjective cognitive decline (Amariglio et al., 2018; Sabatini et al., 2021e; Zimprich & Kurtz, 2015). On the other hand, several studies did not find statistically significant associations for poorer cognitive functioning and cognitive health with subjective cognitive decline (Burmester et al., 2016; Hertzog et al., 2018; Jonker et al., 2000). Hence, it may be that some individuals perceive their cognitive functioning as being poor followed by negative SPA despite having intact objective cognitive functioning.

Second, changes in physical health may also be interpreted by individuals as a consequence of their increased age and may therefore impact their SPA (Kornadt et al., 2019; Spuling et al., 2013; Steverink et al., 2001). Indeed, the occurrence of acute and chronic health problems, and the functional limitations that often accompany them, may increase individuals' general awareness of losses due to aging (Sabatini et al., 2021a; Sargent-Cox et al., 2012; Schönstein et al., 2021; Spuling et al., 2013; Zhang & Neupert, 2020).

Third, and less researched than the previous two domains, poorer mental health, such as increased depressive symptoms, has recently been found as a major predictor of more negative SPA (Schönstein et al., 2021). There may be several reasons for this association. At the action level, for example, depressive symptoms may impair everyday functioning in rather constant ways and, as a consequence, may increase the perception that aging is a negative experience (Kennedy et al., 2007; Stuck et al., 1999). In addition, at the emotional and mood level, the negative self and world views that characterize depressed individuals (Beck, 2002) may detract their attention from the positive experiences that can come with aging. Finally, increased depression may also make aging individuals more susceptible to ageism and may undermine their resilience against negative self-stereotyping (Bodner et al., 2018; Jeste et al., 2013; Wermelinger Ávila et al., 2017).

In conclusion, there is mounting longitudinal evidence suggesting that cognitive functioning and physical and mental health contribute to the formation of older adults' SPA. However, the longitudinal studies exploring the extent to which changes in health influence individuals' SPA are limited in a number of issues. First, research in the area needs a more fine-tuned approach to assess cognitive functioning. Most studies exploring whether cognitive functioning influences cognitive complaints did not assess several cognitive domains. Indeed, they either assessed solely one aspect of cognition (e.g. memory; Hertzog et al., 2018; Seidler & Wolff, 2017) or screened cognition (Burmester et al., 2016) with instruments such as the Mini-Mental State Examination (Folstein et al., 1975). However, cognitive performance essentially is the confluence of numerous abilities such as information processing speed, memory, and inductive reasoning (Hendrie et al., 2006). Exploring the associations of different cognitive abilities with SPA separately is particularly important as different cognitive abilities tend to decline at different rates and magnitude. In some cases, certain cognitive abilities (e.g. general knowledge) may show relatively high stability or may even increase (Christensen, 2001; Deary et al., 2009; Park & Reuter-Lorenz, 2009). Moreover, it could be that different cognitive abilities impact SPA differentially. For instance, decrease in cognitive speed and in working memory may become quite salient in aging individuals' daily life in the contexts of driving or shopping, whereas decline in verbal ability may not be relevant in these contexts (Conlon et al., 2017; Martyr & Clare, 2012; Royall et al., 2007).

Most studies exploring whether physical health influences SPA have used self-rated health as the indicator of a person's physical health (Sabatini et al., 2020a). However, self-rated health does not necessarily match objective levels of health (Schönstein et al., 2021; Staats et al., 1993; Wettstein et al., 2016). Also, Stephan et al. (2015) found that some of the "perceptible" biomarkers (e.g., lower peak expiratory flow, grip strength) in their study were associated with a higher subjective age, whereas imperceptible ones (i.e., blood pressure, telomere length) failed to show such an association. This suggests that individuals may draw on noticeable age-related changes to inform their SPA. Second, most previous work relied on global and unidimensional assessments of SPA. Global measures, such as the single-item felt age indicator (Barrett, 2003), capture an overall evaluation of a person's aging and, therefore, assume that SPA do not vary across different life domains. Additionally, there is an ongoing discussion on whether feeling younger than one's chronological age may be seen as positive or eventually as a variant of ageism (Gendron et al., 2018). Further, important and widely used SPA measures such as the unidimensional Attitudes Toward Own Aging scale (e.g., ATOA scale; Lawton, 1975) treat positive and negative SPA as two ends of the same spectrum and assume that individuals hold either positive or negative SPA, but not both. Given the previous considerations, including available multi-dimensional measures of SPA in

In this regard, a more recently established concept has been the awareness of agerelated change construct (AARC; Diehl & Wahl, 2010), that captures the co-existence of positive (AARC-gains) and negative (AARC-losses) age-related changes. Aside from overall scores for AARC-gains and AARC-losses, the AARC concept offers the assessment of awareness of aging in five different behavioral domains, including health and physical functioning, cognition, interpersonal relationships, social-cognitive and social-emotional functioning, and lifestyle and engagement (Diehl & Wahl, 2010). The perception of AARC-gains is expected to act as a potential facilitator of positive behavior, whereas perceptions of AARC-losses are expected to diminish developmental options in later life. Following tenets of lifespan developmental psychology (Baltes et al., 2006), both AARC-gains and AARC-losses operate in parallel, though not totally independently. This suggests that antecedents for AARC-gains and AARC-losses may be different.

Third, although previous longitudinal studies explored the impact of change in cognitive functioning and health on SPA (Brothers et al., 2020; Sargent-Cox et al., 2012), their observational intervals were limited and hardly longer than 10 years. Also, to the best of our knowledge, no one estimated the role of change in various health indicators that happen in the transition from young-old to very-old age. This transition is among the most critical ones in later life (Baltes & Smith, 2003; Wahl & Ehni, 2020) and is typically associated with a trend toward more negative SPA (Diehl et al., 2021). Still, it may be that long-term and short-term antecedents of SPA in very old age may be different than antecedents in young old age. Information in this regard may be helpful for informing prevention and intervention efforts with older adults.

The Present Study

The current study used data of the ILSE study (<u>https://www.maelstrom-</u> <u>research.org/study/ilse</u>) (see Hildesheim et al., 2019 for overview) and examined whether 20year changes in cognitive functioning, physical, and mental health, that may happen in the transition from young old to very old age, explain variability in levels of AARC-gains and AARC-losses assessed at T4. We also considered the unidimensional ATOA scale for replication and extension purposes. At the independent variable level, we considered major areas of objective cognitive functioning independently. These included working memory, processing speed, perceptual reasoning, spatial ability, general knowledge, and reasoning. Furthermore, we considered indicators of both objective and perceived physical health. Finally, as indicator of mental health, we included self-reported symptoms of depression. We proposed the following hypotheses. First, in terms of objective indicators of cognitive functioning, we expected that abilities with stronger relevance for functioning in everyday life (i.e., information processing speed and working memory) would have larger effects on older adults' SPA than those with lower salience for functioning in everyday life (i.e., verbal ability; Conlon et al., 2017; Martyr & Clare, 2012). We expected this to be especially true for the association of cognitive performance with AARC-losses as these capture perceived decline in everyday life due to getting older. Second, based on findings from previous studies showing that SPA were more strongly associated with self-rated health compared to more objective measures of health (Idler & Benyamini, 1997; Kaspar et al., 2019), we expected that subjective health ratings would show stronger predictive power with regard to participants' SPA compared to objective assessments of physical health. Third, in line with existing evidence (Schönstein et al., 2021), we expected that self-reported depression would predict ATOA, AARC-gains, and AARC-losses.

Fourth, we expected that associations between indicators of cognitive functioning, physical, and mental health and SPA would be significantly stronger for AARC-losses compared to AARC-gains and ATOA for several reasons. First, following tenets of lifespan developmental psychology, loss experiences increasingly outweigh gain-related experiences as people are getting older (Baltes & Smith, 2003; Heckhausen et al., 1989). Hence, increasingly dominating loss experiences in very old age should have a greater influence on AARC-losses compared to AARC-gains. Second, we expected AARClosses to be more strongly related to health indicators compared to ATOA based on the assumption of a matching effect between the experience of losses in the cognitive, physical, and mental domains and the perception of age-related losses in these domains as confirmed by previous SPA studies (Diehl et al., 2021). Third, previous empirical work has shown that the association between AARC and various health-related variables is consistently stronger for AARC-losses compared to AARC-gains (Sabatini et al., 2020a; Sabatini et al., 2021b; Sabatini et al., 2020c).

Methods

Transparency and Openness

This study used data from the ILSE study conducted in Germany (see most recent study description in Hildesheim et al., 2019). The de-identified data on which the study conclusions are based are available to investigators outside the ILSE team after request and approval; requests should be directed to the corresponding author. Contact to the full ILSE data set can be made via one of the co-authors of this paper by means of a spelled-out data agreement (HWW; <u>wahl@nar.uni-heidelberg.de</u>). The analytic code to reproduce analyses is available after request to the corresponding author. The measures used in the ILSE study are publicly available and are described in Hildesheim et al. (2019)

Procedure

In the ILSE study, participants were randomly selected and recruited via city registers from two areas, located in former East and West Germany (Leipzig and Heidelberg). Potential participants were invited to take part in the study via mail. The invitation was followed by a short telephone interview collecting sociodemographic data. Those who agreed to come to the study center located either at the university of Heidelberg or Leipzig underwent a written informed consent procedure.

Participants

The ILSE study baseline assessment (first measurement wave) was conducted between 1993 and 1996, and three measurement waves were conducted 4 (1997/2000), 12 (2005/2008), and 20 (2014/2016) years after baseline assessment. The ILSE study included two age-homogenous subsamples, one born 1950-1952 and the other born 1930-1932. Due to a wider range of cognitive tests being administered only to the 1930-1932 cohort, the current study is based solely on this cohort. The old-age cohort included 500 participants at baseline; 148 participants completed all four measurement waves. Of these 148 participants, 45 individuals were excluded from the current study analyses as they either did not complete the AARC questionnaire (n = 44) or had more than 20% of missing data on the AARC questionnaire (n = 1). The final study sample (see Table 1) included 103 participants (50.5% of women) whose mean age at 20-year follow-up was 82.5 years (SD = 1.0; range: 80-85); mean years of education was 13.5 years. All participants were white Germans representing the German population behind the cohort samples included in ILSE diversity well. At baseline, compared to the study sample, individuals excluded from analyses (n = 397) rated their health as being poorer; scored more poorly on objectively assessed physical health and cognitive tests; were less welleducated; and had more negative ATOA (see Supplemental Online Table S1).

Because individuals who show mild levels of cognitive decline may be among those experiencing and, consequently, perceiving many age-related changes across several life domains (e.g. poorer cognition and mood, reduced social engagement) (Anstey, 2013; Hackett et al., 2019; Hill et al., 2016; Kessler et al., 2012; Mol et al., 2008), we kept those participants who, according to study criteria, were considered to have mild cognitive impairment (MCI; n = 1) at baseline in the study. Research has shown that people with mild cognitive impairment are fairly accurate when evaluating their cognitive performance (Clare et al., 2010).

No participants with a diagnosis of dementia were included. Indication of MCI or dementia were based on a consensus conference where at least two specialists in geriatric psychiatry examined participants' medical and neuropsychological data. In line with the criteria outlined by the International Psychogeriatric Association Working Party (Levy, 1994), MCI was diagnosed when cognitive impairment was reported either by the participant or an informant, or neuropsychological test performance (scores at least one standard deviation below normative levels) indicated objective impairment.

Measures

Dependent Variables: Self-Perceptions of Aging (SPA)

Awareness of Age-Related Change (AARC). To assess AARC-gains and AARC-

losses, we used the AARC 50-item questionnaire (Brothers et al., 2019) which contains 25 items assessing AARC-gains and 25 items assessing AARC-losses. Out of the 50 items, there are 10 items for each of the five AARC life domains. The item stem "*With my increasing age, I notice that*..." precedes either an age-related gain or age-related loss statement. Example items include "... *my physical ability is not what it used to be*" (Health and Physical Functioning-Loss); "... *I am slowed in my thinking*" (Cognitive Functioning-Loss); "... *I appreciate relationships and people much more*" (Interpersonal Relations-Gain); "... *I have a better sense of what is important for me*" (Social-Emotional/Social-Cognitive Functioning-Gain); "... *I have more time for the things I enjoy*" (Lifestyle-Gain). Participants rated how much each item applied to them on a five-point scale (1 = not at all; 5 = very much). Separate scores for AARC-gains and AARC-losses were calculated by summing the 25 items falling into the respective subscales.

Higher scores indicate higher AARC-gains and AARC-losses, respectively, across life domains. For the current study sample, Cronbach's α for AARC-gains was 0.89 and for AARC-losses was 0.91.

Attitude Toward Own Aging (ATOA). We used the 5-item ATOA scale (taken from the Philadelphia Geriatric Center Morale Scale; Lawton, 1975) to assess participants' global SPA. For each statement, respondents were asked to make temporal comparisons about changes in energy level, perceived usefulness, happiness, and quality of life and to respond on a binary response format (*better* versus *worse*, *yes* versus *no*). An example item is "*things keep getting worse as I get older*." A proportion-based score was obtained by summing participant's item scores and by dividing it by the number of responses (possible range: 1-2). Lower scores indicate more negative ATOA, whereas higher scores indicate more positive ATOA. For the current study sample Cronbach's α for ATOA was 0.64.

Independent Variables: Cognitive Functioning, Physical Health, Mental Health

Cognitive Functioning. Cognitive functioning was assessed with a cognitive battery including six paper-pencil tests. The Digit Symbol, Digit Span, and Block Design tests assessed aspects of fluid intelligence, including working memory, processing speed, perceptual reasoning, and spatial ability. The Digit Symbol test (Hoyer et al., 2004; Oswald & Fleischmann, 1995) consists of nine digit-symbol pairs (e.g., 1/V, 2/X..., 9/+) and a list of digits. Participants were given a coding table and were asked to match within 90 seconds as many single digits with corresponding symbols as possible. The number of correctly matched symbols served as the total score (possible range: 0-67).

In the Digit Span test (Oswald & Fleischmann, 1995), participants had to verbally repeat a series of single digits (e.g., "5, 6, 7") either in the same order as they were presented by the interviewer (Part 1) or in reversed order (Digit Span backwards, Part 2). The number of digits increased by one at each trial until the participant consecutively failed two trials of the same length. Test scores were calculated by summing correct answers (possible range: 0-8).

The Block Design test was taken from the German version of the WAIS-R (Tewes, 1991; Wechsler, 1981). Participants were required to rearrange four or nine block designs according to abstract patterns. Both accuracy and time taken to complete the four or nine items were incorporated into the scoring of the test (possible range: 0-51).

The Information, Similarities, and Picture Completion tests assessed crystallized intelligence and were taken from the German version of the WAIS-R (Tewes, 1991; Wechsler, 1981). For the Information test, participants answered 24 questions on general knowledge topics (e.g., "*What is the capital of Turkey*?"). Every correct response was scored with 1 point (possible range: 0-24). In the Similarities test, participants were presented with 16 pairs of words and asked how the words in each pair were alike (e.g., egg-seed). Answers to each pair of words were rated with 0, 1, or 2 points, depending on the quality of the response, and summed up to obtain a total score (possible range: 0-32). In the Picture Completion test, participants were shown pictures with an important part missing (e.g., frog with only three legs) and were asked to identify the object of the picture. Correct answers were scored with 1 point and summed up to obtain a total score (possible range: 0-17). For each of the six cognitive tests higher scores indicated better performance.

Objectively Assessed Physical Health. To assess physical health a trained study physician conducted an in-depth clinical examination, comprised of a medical check-up, hearing

and vision assessment, a laboratory blood test, and a geriatric assessment (see Miche et al., 2014a). At the end of the examination participants' health was rated on a 6-point scale (1 = "Participant exhibits a serious medical condition, which is immediately life-threatening; professional healthcare is urgently needed;" 6 = "Participant exhibits very good health;" i.e., no chronic disease, no chronic pain, all clinical assessments suggest nonpathological findings).

Subjective Health Rating. A one-item question asked participants to evaluate their health status on a 6-point scale (1 = *very bad*; 6 = *very good*) (Ware & Sherbourne, 1992).

Depressive Symptoms. Depressive symptoms were assessed with the 20-item Self-Rating Depression Scale (Zung, 1965). For each item, respondents indicated how frequently they experienced the symptom (1 = never; 4 = always). The mean of the total score (possible range: 1-4) was calculated by dividing the sum of single-item scores by the total number of items; higher scores indicated more severe depressive symptoms. Psychometric properties of the scale are good among older adults (Jokelainen et al., 2019).

Measures were administered at all four ILSE measurement waves with the exception of the AARC questionnaire which, due to being a more recently developed questionnaire, was administered only during the fourth ILSE measurement wave.

Covariates

Baseline sex, education, and depressive symptoms were used as covariates. Baseline depression was used as an additional covariate when exploring indicators of cognitive functioning and physical health. Education was assessed with a categorical variable including four answer options (1 = up to 10 years of education; 2 = between 11 and 12 years; 3 = between 13 and 15 years; 4 = more than 15 years of school and university education). These variables were used as covariates because SPA vary among men and women (Sabatini et al., 2021c), among individuals with different educational achievements (Sabatini et al., 2020b), and in relation to levels of depressive symptoms (Dutt et al., 2016a). Moreover, prevalence of cognitive and physical health decline differs between men and women and between individuals with different educational achievements (Anstey, 2013; World Health Organization, 2018).

Statistical Analyses

To maximize the use of data, mean imputation based on the group mean was computed when up to 20% of responses was missing in the ATOA scale; we did so for 10 participants. No missing data were present in the items of the remaining measures.

To explore between- and within-person 20-year change in the six cognitive tests, objectively assessed physical health, subjective health rating, and depressive symptoms, we subtracted the 20-year follow-up scores from the baseline scores. Hence, positive scores indicate a decline in cognitive functioning and physical health, whereas negative scores indicate an increase in depressive symptoms. We used matched paired t-test analyses to test the statistical significance of change between baseline and follow-up score in each of the nine health indicators. Descriptive analyses were conducted for main study variables and Pearson's r correlation coefficients were calculated among main study variables.

To estimate the individual predictive role of 20-year change in participants' performance in cognitive functioning, objectively assessed physical health, subjective health, and depressive symptoms on levels of AARC-gains, AARC-losses, and ATOA, we performed linear regression analyses. We checked the linearity of change in the health indicators over the study period (see Figures 1-9). We adopted an exploratory approach where we first conducted nine univariate linear regressions with each of the nine indicators of 20-year change as the predictor and AARC-gains, AARC-losses, or ATOA as the outcome. For those indicators of change in health and cognitive functioning that were statistically significant predictors of SPA, we performed a second model adjusting for demographic variables (sex and education) and a third model adjusting for demographic and health-related variables. Health-related covariates varied depending on the predictive variable assessed. We adjusted for subjective health rating and depressive symptoms when exploring the predictive role of change in cognition; for depressive symptoms when exploring the predictive role of change in objectively assessed physical health and subjective health rating; and for subjective health rating when exploring the predictive role of change in depression coefficients are reported to indicate the size of effects; we considered coefficients $\leq .09$ as negligible, .10-.29 as small, .30-.49 as moderate and $\geq .50$ as large (Cohen, 1988). Significance level was set at .05.

To further explore between- and within-person variation in baseline levels of the nine health indicators, as well as in the rate of change of these indicators over 20 years (across four assessment waves), we fit latent growth curve models (LGCM). Compared to more classical ways of analyzing change over time, such as repeated-measures analysis of variance (R-ANOVA), LGCM have several advantages, including the incorporation of latent variables, the testing of different trajectories of change, and comparison of change across sub-groups (Preacher et al., 2008).

For each of the nine health indicators, we estimated four models. The first model (fixed intercept models) estimated average baseline level of the selected health indicator.

The second model (random intercept model) estimated both average baseline level of the selected health indicator and variation between individuals in the baseline level of the selected health indicator. The third model (random intercept, fixed slope model) estimated average baseline level of the selected health indicator, variation between individuals in the baseline level of the selected health indicator, and average linear change in the selected health indicator over 20 years. The fourth model (random intercept and random slope model) estimated average baseline level of the selected health indicator, variation between individuals in the baseline level of the selected health indicator, variation between individuals in the baseline level of the selected health indicator, variation between individuals in the baseline level of the selected health indicator, average linear change in the selected health indicator over 20 years, and between-person variation in the rate of change of the selected health indicator over 20 years.

In the models we took into account that the time interval between the first and second follow-up and between the second and third follow-up was twice as large than between baseline assessment and the first follow-up. As indicators of model fit we reported the Comparative Fit index (CFI), the Root Mean Square Error of Approximation (RMSEA), and the Standardized Root Mean Square Residual (SRMR). Generally, CFI > .90, RMSEA < .08 (95% CI: between 0 and .08), and SRMR < .06 indicate adequate model fit (Byrne, 2012). Power analysis for LGCM was conducted with an online sample size calculator for structural equation models (https://www.danielsoper.com/statcalc/calculator.aspx?id=89). Estimations suggested that for a model with power = .80, effect size = 0.3 (as found in previous research; Siebert et al., 2016), and probability level = 0.05, a sample of minimum 90 participants was needed.

Finally, although the main interest of our study was to explore predictors of SPA, we acknowledge that the associations between health indicators and SPA may be bidirectional. We therefore estimated cross-lagged panel models (CLPM) to explore the bidirectional influence between ATOA and those health indicators that resulted to be statistically significant predictors of ATOA. It was not possible to explore the bidirectional associations of health indicators with AARC-gains and AARC-losses as these were assessed only at 20-year follow-up. Each model consisted of two predictors (baseline scores on ATOA and on the selected health indicator) and two outcomes (20-year follow-up scores on ATOA and on the selected health indicator). In these models, ATOA was allowed to correlate with the selected health indicator. Each model estimated two autoregressive paths, representing how stable scores on ATOA and on the selected health indicator were over 20 years, and two cross-lagged paths, representing the extent to which baseline ATOA influenced follow-up scores in the selected health indicator and the extent to which baseline scores in the selected health indicator influenced follow-up levels of ATOA.

Except from LGCMs that were estimated using Mplus software (Muthén & Muthén, 2017), all remaining study analyses were conducted using STATA version 16 (StataCorp, 2017).

Results

Descriptive Analyses

Descriptive statistics for the study sample are reported in Table 1. On average, at 20-year follow-up participants reported moderate levels of AARC-gains (M = 78.31; SD = 13.79; range: 33-115; theoretical range: 25-125) and few AARC-losses (M = 61.02; SD = 14.37; range: 28-94; theoretical range: 25-125), and positive ATOA (M = 1.57; SD = 0.31; range: 1-2). Mean scores for AARC-gains and AARC-losses across the five AARC life domains are reported in Supplemental Online Table S2. Over 20 years, participants' performance in the Digit Span, Digit Symbol, Similarities, Block Design, and Picture Completion tests declined, whereas participants' performance in the Information test improved; objectively assessed physical health and

subjective health rating decreased; and depressive symptoms increased (see Table 2). Supplemental Online Figures 1-9 and Supplemental Online Table S3 show between- and withinperson change in each of the nine health indicators over 20 years. The health and cognitive predictors with greater variability in the rate of change between persons were the Information and Block Design tests, objectively assessed physical health, and depressive symptoms. Although for the majority of participants scores on the Block Design test (see Figure 5) and on objectively assessed physical health (see Figure 7) declined, for some participants scores on these tests improved or stayed stable. Similarly, for the majority of participants scores on the Information test (see Figure 3) and depressive symptoms (see Figure 9) increased, but for a minority of participants performance on the Information test and depressive symptoms declined.

At 20-year follow-up more AARC-losses were correlated with more negative ATOA (r = -.58, p < .001), whereas the correlations between AARC-gains and ATOA (r = -.12, p = .22), and between AARC-gains and AARC-losses (r = -.12, p = .24) were not statistically significant. Correlations among follow-up levels of global and domain specific AARC-gains and AARC-losses and ATOA with baseline and follow-up scores on the nine cognitive and health indicators are reported in Table 3 and Supplemental Online Table S4. Most correlations were of either small or negligible size. Only the correlations for AARC-losses with subjective health rating and depressive symptoms were of moderate size.

Change in Cognitive Functioning, Physical and Mental Health as Predictor of AARC-Gains

From the univariate linear regressions (Table 4) with each indicator of 20-year change in cognitive functioning, physical, and mental health predicting AARC-gains, only a decline in participants' performance on the Digit Symbol test predicted fewer AARC-gains. The small size

of the effect ($R^2 = .07$) remained statistically significant after controlling for the effects of demographic variables, subjective health rating, and depressive symptoms. Changes in any of the other cognitive tests failed to be significant predictors of AARC-gains.

Change in Cognitive Functioning, Physical and Mental Health as Predictor of AARC-Losses

From the univariate linear regressions (Table 4) with each indicator of 20-year change in cognitive functioning, physical and mental health predicting AARC-losses, poorer subjective health rating, greater decline in participants' performance on the Digit Symbol and Picture Completion tests, and an increase in depressive symptoms predicted greater AARC-losses. The association between decline in Digit Symbol and fewer AARC-gains is similar in size to that between decline in Digit Symbol and higher AARC-losses. Changes in the other cognitive tests as well as changes in objective health were not significant predictors of AARC-losses.

Results of multiple regressions with indicators of change in health as predictors of AARC-losses are reported in Table 5. Decline in participants' performance in the picture completion test was no longer a statistically significant predictor of AARC-losses after adjusting for demographic variables. The remaining associations remained statistically significant after adjusting for covariates.

Change in Cognitive Functioning, Physical and Mental Health as Predictor of ATOA

From the univariate linear regressions (Table 4) with each indicator of 20-year change in cognitive functioning, physical and mental health predicting ATOA, a decline in participants' subjective health rating and an increase in depressive symptoms were significant predictors of more negative ATOA. Changes in the cognitive tests were not significant predictors of ATOA.

The moderate relevance of a decline in subjective health and an increase in depressive symptoms for predicting negative ATOA remained in the full adjusted models (See Table 5).

Bidirectional Associations of ATOA with Physical and Mental Health

Results from the cross-lagged panel models exploring the bidirectional associations of ATOA with subjective health rating and depressive symptoms are reported in Supplemental Online Table S5. The autoregressive paths show that baseline levels of ATOA predicted variance in 20-year follow-up ATOA and baseline scores on health indicators predicted variance in 20-year follow-up scores on the same health indicator. Baseline subjective health rating and ATOA did not predict each other at 20-year follow-up. Baseline depressive symptoms and ATOA significantly predicted each other at 20-year follow-up.

Discussion

This study explored whether 20-year changes in cognitive functioning, objective physical health, subjectively rated health, and depressive symptoms predict levels of AARC-gains, AARC-losses, and ATOA in a sample of very old adults aged 80 to 85 years. We proposed four hypotheses: (1) Change in objective indicators of cognitive functioning with strong associations to cognitive functioning in daily life will have larger effects on SPA than those less salient in everyday life. (2) Subjectively rated health will be a stronger predictor of SPA than objectively assessed physical health. (3) Depressive symptoms will emerge as a significant predictor of SPA. (4) The associations between different predictors and SPA will be stronger for AARC-losses compared to AARC-gains and ATOA.

We found partial support for the first hypothesis as a decline in participants' performance in the Digit Symbol test, assessing information processing speed, predicted fewer AARC-gains and more AARC-losses. However, despite the fact that working memory becomes particularly salient in very old age with regard to everyday tasks such as driving or shopping (Conlon et al., 2017; Jonker et al., 2000; Martyr & Clare, 2012; Royall et al., 2007), we did not find a significant association between average decline in working memory and greater AARC-losses. We confirmed the second hypothesis showing that a decrease in subjectively rated health, but not in objectively assessed physical health, significantly predicted greater AARC-losses and more negative ATOA. In line with our third hypothesis, an increase in depressive symptoms was a significant predictor of greater AARC-losses and more negative ATOA, but not of lower AARCgains. Finally, we found partial support for the fourth hypothesis because only some associations between the investigated predictors and SPA were stronger for AARC-losses compared to AARC-gains and ATOA. Specifically, subjectively rated health and depressive symptoms were more strongly associated with AARC-losses compared to AARC-gains, whereas the indicators of cognitive functioning were similarly associated with both AARC-gains and AARC-losses. Decline in the Digit Symbol test was more strongly associated with AARC-losses compared to ATOA, whereas subjective and objective health and depressive symptoms were similarly associated with both AARC-losses and ATOA.

Cognitive Functioning as Predictor of SPA

The small association found between a decline in participants' information processing speed and greater AARC-losses is in line with cross-sectional evidence, documenting either negligible or small associations between indicators of poorer subjective and objective cognitive performance and greater AARC-losses (O'Brien et al., 2020; Sabatini et al., 2020b; Sabatini et al., 2021b; Zhu & Neupert, 2020). Differently from cross-sectional evidence showing that poorer cognitive performance is negligibly associated with more AARC-gains (Sabatini et al., 2021b), the current longitudinal study found a small association between decline in cognitive functioning

and lower AARC-gains. The discrepancy between cross-sectional and longitudinal evidence highlights the importance of linking AARC-gains and AARC-losses with the assessment of change in developmental outcomes, such as cognitive functioning. Indeed, AARC-gains and AARC-losses respectively very likely capture a trajectory of perceived cognitive improvement or decline rather than perceptions of current cognitive functioning.

Although on average participants' performance on the Digit Span, Block Design, Information, Similarities, and Picture Completion tests declined significantly over the study period, change in these cognitive domains did not predict levels of AARC-gains and AARClosses. However, given the great between-person variability found in the rate of change of working memory (Digit Span) and knowledge (Information), it may be that our analyses linking average change in working memory and knowledge to AARC-gains and AARC-losses failed to detect significant associations among those who experienced grater decline in working memory and knowledge. Nonetheless, study results suggest that only change in information processing may be a risk factor for the perception of fewer AARC-gains and greater AARC-losses. Individuals reporting low AARC-gains and high AARC-losses may, therefore, benefit from further cognitive monitoring and from interventions aiming to delay or prevent the progress of age-related cognitive decline to levels of pathological cognitive decline (Hahn & Lachman, 2015; Rakesh et al., 2017).

Change in cognition did not predict ATOA. However, previous longitudinal evidence, including a study using data from the ILSE study, shows that more positive ATOA predicts better performance on cognitive tests (Levy et al., 2018; Robertson & Kenny, 2016; Seidler & Wolff, 2017; Siebert et al., 2020; Siebert et al., 2016). Hence, although ATOA is a wellestablished predictor of future cognitive ability, individuals' ATOA may not be influenced by objectively observed cognitive decline. An explanation for decline in cognitive performance being a significant predictor of AARC-losses but not of ATOA may be due to AARC capturing dynamic self-evaluations of cognition rooted in a person's everyday life (Miche et al., 2014b), whereas the ATOA scale assesses more global and overall perceptions of aging that are not necessarily linked to specific experiences (Hess, 2006; Lawton, 1975; Miche et al., 2014a).

Physical Health as Predictor of SPA

The finding that a decline in subjectively rated health predicted greater AARC-losses and more negative ATOA extends previous evidence showing that more negative subjective health ratings are a longitudinal predictor of more negative scores on other measures of SPA, such as an older felt age (Spuling et al., 2013). The lack of a significant association between decline in subjective health and fewer AARC-gains is in line with cross-sectional evidence showing that the association between better physical health indicators and greater AARC-gains tends to be negligible (Brothers et al., 2017; Sabatini et al., 2020a; Sabatini et al., 2020c). This result may be due to the fact that physical losses can still be apparent for a person who reports many AARC-gains, given that AARC-gains and AARC-losses tend be positively correlated (Sabatini et al., 2020d).

The lack of a significant association between a decline in objectively assessed physical health and SPA is not in line with cross-sectional evidence, supporting the associations of poorer physical health with fewer AARC-gains, more AARC-losses (Dutt et al., 2018a), and less positive ATOA (Westerhof et al., 2014). Our results may however support the "health paradox," stating that even though physical and functional health generally decline with aging, self-perceptions of health remain positive or change to a lesser extent (Idler & Benyamini, 1997;

Jylha et al., 2001). However, this is unlikely as in our study, assessing five AARC life domains, participants perceived the highest levels of losses in the health and physical functioning domain. Alternatively, the non-significant association between participants' average decline in objectively assessed physical health and AARC at follow-up may be due to participants experiencing, on average, only a minimal decline in their objective physical health. However, because we found great variance in the trajectory of objective physical health between participants, it may be that those few participants who experienced greater decline in their objective physical health perceived higher AARC-losses at follow-up. Future studies should explore this hypothesis as our sample size prevented us from testing it in the current study.

Although regression analyses showed that change in objectively assessed physical health did not predict follow-up levels of ATOA, previous longitudinal evidence showed that the presence of chronic health conditions at baseline predicted more negative SPA (Schönstein et al., 2021). Overall, it may be that although baseline objective assessments of physical health are informative of future levels of SPA, the assessment of a decline in objectively assessed physical health is not always informative of future levels of SPA. It may well be that small changes in objective health, as those experienced by the current study sample, are not noticed by individuals and that only the occurrence of severe changes in objective health leads to an increase in individuals' awareness of age-related losses and to more negative view of aging.

Mental Health as Predictor of SPA

Across the study period, participants reported a significant increase in depressive symptoms. Although an increase in depressive symptoms may be more likely when transitioning from young-old to very-old age, due to normative loss experiences, those individuals who experienced a more severe increase in depressive symptoms were more vulnerable to the experience of AARC-losses and negative ATOA. This pattern of results was expected as in older age depressive symptoms tend to go along with poor physical and functional health (Alexopoulos, 2005; Fauth et al., 2012), and awareness of physical and functional decline are assessed with items in the AARC-losses and ATOA subscale. The predictive role of change in depressive symptoms over more negative ATOA was in line with recent evidence (Schönstein et al., 2021). The findings of this study suggest a bidirectional influence between more negative ATOA and more depressive symptoms. Although in this study it was not possible to explore the bidirectional relationship between depressive symptoms and AARC-losses, due to AARC-losses having been assessed only at follow-up, there may be a bidirectional relationship also between AARC-losses and depression. Indeed, we found that an increase in depressive symptoms significantly predicted more AARC-losses at 20-year follow-up and longitudinal evidence showed that more AARC-losses at baseline predicted an increase in depressive symptoms (Dutt et al., 2016a). Overall, addressing depressive symptoms in older age may be a way of preventing and/or limiting negative SPA. At the same time, the emotional well-being of individuals reporting high AARC-losses and negative ATOA may be enhanced with psychological interventions promoting acceptance of negative age-related changes and ego-integrity (Collins & Kishita, 2018; Vailati Riboni et al., 2018; Vailati Riboni et al., 2020; Zanini et al., 2018).

Cross-sectional evidence on the association of depression with AARC-gains has been inconsistent (Sabatini et al., 2020a). Some studies reported a significant association between fewer depressive symptoms and more AARC-gains (Kaspar et al., 2019), whereas others did not (Dutt et al., 2016a; Dutt & Wahl, 2018; Dutt et al., 2018b). The findings from this longitudinal study suggest that a more severe increase in depressive symptoms does not necessarily predict lower AARC-gains. It may be that even though in the transition from young-old to very-old age an increase in depressive symptoms is more likely due to the experience of age-related losses (Alexopoulos, 2005; Palsson et al., 2001; Weyerer et al., 2013), loss-related depressive symptoms may not necessarily prevent individuals from having experiences and perceptions of age-related gains.

Although developmental theories postulate that in very old age losses tend to outweigh gains (Baltes, 1997), in the current study, participants reported more positive than negative ATOA, moderate levels of AARC-gains and only few AARC-losses. These levels of SPA were similar to those found in other cohort studies of community-dwelling older adults (Brothers et al., 2017; Sabatini et al., 2021b). Positive SPA among community-dwelling older individuals have also been reported by studies investigating other indicators of SPA such as felt age (Westerhof et al., 2003). The overly positive SPA reported in this and other cohort studies may be due to most samples comprising self-selective groups of older individuals who experience less than average decline in cognitive functioning, physical, and mental health. It may be that when very old individuals enjoy reasonably good health, they do not need to shift their resources away from growth and, as a consequence, positive age-related experiences and perceptions may still outnumber negative ones (Baltes, 1997).

In sum, by exploring, for the first time, changes in cognitive functioning as predictors of AARC-gains and AARC-losses, this study made it possible to extent previous cross-sectional and micro-longitudinal evidence on the associations of AARC-gains and AARC-losses with cognitive functioning (Sabatini et al., 2021b; Zhu & Neupert, 2020). We confirmed that the association between poorer cognition and more AARC-losses is consistently small also at the longitudinal level. Hence, in addition to cognition, many psychological variables, such as mood,

self-efficacy, and beliefs about age-related cognitive changes, may shape adults' levels of AARC-losses (Dutt & Wahl, 2018; Sabatini et al., 2021b).

Moreover, the exploration of changes in mental and physical health as predictors of AARC-gains and AARC-losses made it possible to complement previous knowledge on the predictive role of higher AARC-gains and lower AARC-losses over better mental and physical health (Brothers et al., 2020). Indeed, as we found that AARC-losses were predicted by a decline in subjective health ratings and an increase in depressive symptoms, AARC-losses may not only be a predictor of poorer future mental and physical health (Brothers et al., 2020) but also be the outcome of decline in mental and physical health.

Although both AARC-losses and more negative ATOA were predicted by an increase in depressive symptoms and poorer subjective health, AARC-gains and AARC-losses were not predicted by change in objectively assessed physical health. This pattern of results may be due to AARC-losses and ATOA being measured with similar methods to those used to assess subjective health and depressive symptoms (i.e., self-reports) and with different methods compared to those used to assess cognitive functioning and objective physical health (i.e., behavioral measures of cognitive functioning and objective physical examination). Nonetheless, both the AARC-losses and ATOA subscales may be useful to identify those individuals who experience depressive symptoms and perceive their health as being poor but may not be useful to identify those individuals with poorer objective health. These individuals may benefit from intervention programs promoting health and more positive and realistic views of aging (Brothers & Diehl, 2017; Levy et al., 2014; Menkin et al., 2020; Sarkisian et al., 2007; Wolff et al., 2014).

Strengths and Limitations

This study has several strengths. First, the 20-year longitudinal study design is a strength because most available evidence to date on the association of AARC with health indicators comes from cross-sectional studies (Sabatini et al., 2020a) or relies on studies with shorter follow-up periods (Brothers et al., 2020; Dutt et al., 2018b). Moreover, the longitudinal study design made it possible to discover that change in health predicts AARC-gains and AARC-losses, providing empirical support for the value of the framework of AARC (Diehl & Wahl, 2010). Second, in contrast to most evidence that explored the association of AARC with physical health relying on ratings of subjective health, this study used both a subjective and an objective assessment of physical health. This is a strength as subjective and objective ratings of physical health are not interchangeable and often show divergent associations with outcomes (Althubaiti, 2016; Schönstein et al., 2021; Staats et al., 1993).

Third, because in ILSE the cognitive tests were administered by a trained researcher, it was possible to observe participants' behavior and engagement while taking the cognitive tests. Thus, participants' cognitive test scores were derived directly and were not based on proxy reports.

Fourth, as participants' change in cognitive functioning was assessed across six different domains, it was possible to take an exploratory approach and identify which aspects of cognition showed the strongest associations with individuals' SPA. Moreover, the comprehensive cognitive battery used in the current study provided a more accurate assessment of participants' cognitive functioning compared to that of similar studies that relied on simple screening measures of cognition (Schönstein et al., 2021).

This study has also several limitations. First, similarly to most studies collecting information about health and lifestyle among older individuals, the participants that were

included in the current analyses, may have been a self-selected group of individuals with above average health and education compared to the broader German population of very old individuals (European Union, 2019; Gerstorf et al., 2006; Klaus et al., 2017; Puth et al., 2017). Hence, because the sample very likely included individuals who had aged more successfully than the average older adult, the association we found for a decline in cognition and subjective health and an increase in depressive symptoms with more negative SPA, may be stronger when investigated in a more heterogeneous sample with greater variability in these key variables. Second, although in the ILSE study several years passed between one assessment and the next, the administration of the same cognitive tests at baseline and at three subsequent follow-ups may have given rise to some practice effects (Goldberg et al., 2015) and may have obscured the magnitude of cognitive decline experienced by participants. However, a practice effect may be expected in cognitively healthy older individuals and lack of practice effect may indicate some sort of pathological cognitive decline (Duff et al., 2010; Duff et al., 2007).

Third, due to the AARC questionnaire having been recently developed, a limitation of this study is that AARC-gains and AARC-losses were assessed only at 20-year follow-up. As a consequence, it was not possible to examine change in AARC nor whether change in AARC was related to changes in cognitive functioning, physical health, and mental health. Moreover, although we explored the bidirectional association of ATOA with health indicators, it was not possible to test the bidirectional associations between AARC (gains and losses) and health indicators. We therefore refrain from causal inferences on the relationship of AARC-gains and AARC-losses with cognitive functioning, physical health, and mental health. Fourth, exploring the associations of SPA with nine indicators of change in health may introduce a problem of multiplicity. However, the aim of study was to explore several indicators of change in cognitive functioning and health as a way of identifying a potential group of individuals at high-risk of negative SPA. Fifth, as there were many missing cases for objectively assessed physical health, estimates for the associations of change in objectively assessed physical health and SPA outcomes were less reliable. Sixth, although in Supplemental Online Material we report results for LGCMs exploring the rate of change in the health indicators over the study period, the limited statistical power of the current study prevented us from exploring AARC-gains, AARC-losses, and ATOA as outcomes in these LGCMs. Seventh, to explore the bidirectional associations of ATOA with health indicators we used CLPM (Selig & Little, 2012) although their use is currently being debated.

Conclusions

This study explored the predictive associations of 20-year change in objective cognitive performance, objective and subjective physical health, and depressive symptoms with levels of AARC-gains, AARC-losses, and ATOA. We found that the more individuals declined in their performance on the cognitive test "Digit Symbols", perceived a decline in their physical health, and experienced an increase in their depressive symptoms, the more they were also likely to report negative SPA. Hence, more negative SPA may be useful indicators that may help to detect those individuals with poorer health and who would, therefore, benefit the most from health-promoting interventions.

References

- Alexopoulos, G. S. (2005). Depression in the elderly. *The Lancet, 365*(9475), 1961-1970. https://doi.org/10.1016/S0140-6736(05)66665-2
- Althubaiti, A. (2016). Information bias in health research: Definition, pitfalls, and adjustment methods. *Journal of Multidisciplinary Healthcare*, 9, 211-217. https://doi.org/10.2147/JMDH.S104807
- Amariglio, R. E., Buckley, R. F., Mormino, E. C., Marshall, G. A., Johnson, K. A., Rentz, D. M., & Sperling, R. A. (2018). Amyloid-associated increases in longitudinal report of subjective cognitive complaints. *Alzheimer's & Dementia*, *4*, 444-449.

https://doi.org/10.1016/j.trci.2018.08.005

- Anstey, K. J. (2013). Optimizing cognitive development over the life course and preventing cognitive decline: Introducing the Cognitive Health Environment Life Course Model (CHELM). *International Journal of Behavioral Development, 38*(1), 1-10.
 https://doi.org/10.1177/0165025413512255
- Baltes, P. B. (1997). On the incomplete architecture of human ontogeny: Selection, optimization, and compensation as foundation of developmental theory. *American Psychologist*, 52(4), 366-380. <u>https://doi.org/10.1037/0003-066X.52.4.366</u>
- Baltes, P. B., & Smith, J. (2003). New frontiers in the future of aging: From successful aging of the young old to the dilemmas of the fourth age. *Gerontology*, 49(2), 123-135. <u>https://doi.org/10.1159/000067946</u>
- Barrett, A. E. (2003). Socioeconomic status and age identity: The role of dimensions of health in the subjective construction of age. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences, 58*(2), 101-109. <u>https://doi.org/10.1093/geronb/58.2.S101</u>

- Beck, A. T. (2002). Cognitive models of depression. In R. L. Leathy & E. T. Dowd (Eds.), *Clinical Advances in Cognitive Psychotherapy: Theory and Application* (Vol. 14, pp. 29-61). Springer Publishing Co.
- Bodner, E., Palgi, Y., & Wyman, M. F. (2018). Ageism in mental health assessment and treatment of older adults. In L. Ayalon & C. Tesch-Römer (Eds.), *Contemporary Perspectives on Ageism* (pp. 241-262). Springer International Publishing.
 https://doi.org/10.1007/978-3-319-73820-8_15
- Brothers, A. F., & Diehl, M. K. (2017). Feasibility and efficacy of the Aging(Plus) program: Changing views on aging to increase physical activity. *Journal of Aging and Physical Activity*, 25(3), 402-411. <u>https://doi.org/10.1123/japa.2016-0039</u>
- Brothers, A. F., Gabrian, M., Wahl, H.-W., & Diehl, M. K. (2019). A new multidimensional questionnaire to assess awareness of age-related change (AARC). *The Gerontologist*, 59(3), e141-e151. <u>https://doi.org/10.1093/geront/gny006</u>
- Brothers, A. F., Kornadt, A. E., Nehrkorn-Bailey, A., Wahl, H.-W., & Diehl, M. K. (2020). The effects of age stereotypes on physical and mental health are mediated by self-perceptions of aging. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences, 76*(5), 845-857. <u>https://doi.org/10.1093/geronb/gbaa176</u>
- Brothers, A. F., Miche, M., Wahl, H.-W., & Diehl, M. K. (2017). Examination of associations among three distinct subjective aging constructs and their relevance for predicting developmental correlates. *The Journals of Gerontology, Series B: Psychological Sciences* and Social Sciences, 72(4), 547-560. <u>https://doi.org/10.1093/geronb/gbv085</u>
- Burmester, B., Leathem, J., & Merrick, P. (2016). Subjective cognitive complaints and objective cognitive function in aging: A systematic review and meta-analysis of recent cross-

sectional findings. Neuropsychology Review, 26(4), 376-393.

https://doi.org/10.1007/s11065-016-9332-2

- Byrne, B. M. (2012). Structural equation modeling with Mplus: Basic concepts, applications, and programming. Routledge/Taylor & Francis Group.
- Christensen, H. (2001). What cognitive changes can be expected with normal ageing? *Australian* and New Zealand Journal of Psychiatry, 35(6), 768-775. <u>https://doi.org/10.1046/j.1440-</u> <u>1614.2001.00966.x</u>
- Clare, L., Whitaker, C. J., & Nelis, S. M. (2010). Appraisal of memory functioning and memory performance in healthy ageing and early-stage Alzheimer's Disease. *Aging, Neuropsychology, and Cognition, 17*(4), 462-491. https://doi.org/10.1080/13825580903581558
- Cohen, J. (1988). Set correlation and contingency tables. *Applied Psychological Measurement*, *12*(4), 425-434. <u>https://doi.org/10.1177/014662168801200410</u>
- Collins, R. N., & Kishita, N. (2018). The effectiveness of mindfulness- and acceptance-based interventions for informal caregivers of people with dementia: A meta-analysis. *The Gerontologist*, *59*(4), e363-e379. https://doi.org/10.1093/geront/gny024
- Conlon, E. G., Rahaley, N., & Davis, J. (2017). The influence of age-related health difficulties and attitudes toward driving on driving self-regulation in the baby boomer and older adult generations. Accident Analysis & Prevention, 102, 12-22. https://doi.org/10.1016/j.aap.2017.02.010
- Deary, I. J., Corley, J., Gow, A. J., Harris, S. E., Houlihan, L. M., Marioni, R. E., Penke, L., Rafnsson, S. B., & Starr, J. M. (2009). Age-associated cognitive decline. *British Medical Bulletin*, 92, 135-152. <u>https://doi.org/10.1093/bmb/ldp033</u>

- Diehl, M. K., Brothers, A. F., & Wahl, H.-W. (2021). Self-perceptions and awareness of aging:Past, present, and future. In K. W. Schaie & S. L. Willis (Eds.), *Handbook of the psychology of aging* (8th ed.). Elsevier.
- Diehl, M. K., & Wahl, H.-W. (2010). Awareness of age-related change: Examination of a (mostly) unexplored concept. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences*, 65B(3), S340-S350. <u>https://doi.org/10.1093/geronb/gbp110</u>
- Duff, K., Beglinger, L. J., Moser, D. J., Paulsen, J. S., Schultz, S. K., & Arndt, S. (2010).
 Predicting cognitive change in older adults: The relative contribution of practice effects.
 Archives of Clinical Neuropsychology, 25(2), 81-88.

https://doi.org/10.1093/arclin/acp105

- Duff, K., Beglinger, L. J., Schultz, S. K., Moser, D. J., McCaffrey, R. J., Haase, R. F., Westervelt, H. J. K., Langbehn, D. R., Paulsen, J. S., & Huntington's Study Group.
 (2007). Practice effects in the prediction of long-term cognitive outcome in three patient samples: A novel prognostic index. *Archives of Clinical Neuropsychology*, 22(1), 15-24. https://doi.org/10.1016/j.acn.2006.08.013
- Dutt, A. J., Gabrian, M., & Wahl, H.-W. (2016a). Awareness of age-related change and depressive symptoms in middle and late adulthood: Longitudinal associations and the role of self-regulation and calendar age. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences, 73*(6), 944-953. https://doi.org/10.1093/geronb/gbw095
- Dutt, A. J., Gabrian, M., & Wahl, H.-W. (2016b). Developmental regulation and awareness of age-related change: A (mostly) unexplored connection. *The Journals of Gerontology,*

Series B: Psychological Sciences and Social Sciences, 73(6), 934-943.

https://doi.org/10.1093/geronb/gbw084

- Dutt, A. J., & Wahl, H.-W. (2018). Future time perspective and general self-efficacy mediate the association between awareness of age-related losses and depressive symptoms. *European Journal of Ageing*, 16(2), 227-236. <u>https://doi.org/10.1007/s10433-018-0482-3</u>
- Dutt, A. J., Wahl, H.-W., & Diehl, M. K. (2018a). Awareness of aging processes. In B. Knight(Ed.), Oxford Research Encyclopedia of Psychology and Aging. Oxford University Press.
- Dutt, A. J., Wahl, H.-W., & Rupprecht, F. (2018b). Mindful vs. mind full: Processing strategies moderate the association between subjective aging experiences and depressive symptoms. *Psychology and Aging*, 33, 630-642. <u>https://doi.org/10.1037/pag0000245</u>

European Union. (2019). Ageing Europe- Looking at the lives of older people in the EU.

Fauth, E. B., Gerstorf, D., Ram, N., & Malmberg, B. (2012). Changes in depressive symptoms in the context of disablement processes: Role of demographic characteristics, cognitive function, health, and social support. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences, 67*(2), 167-177.

https://doi.org/10.1093/geronb/gbr078

- Folstein, M. F., Folstein, S. E., & McHugh, P. R. (1975). "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. *Journal of Psychiatric Research*, 12(3), 189-198. <u>https://doi.org/10.1016/0022-3956(75)90026-6</u>
- Gendron, T. L., Inker, J., & Welleford, A. (2018). "How old do you feel?" The difficulties and ethics of operationalizing subjective age. *The Gerontologist*, 58(4), 618-624. <u>https://doi.org/10.1093/geront/gnx098</u>

- Gerstorf, D., Herlitz, A., & Smith, J. (2006). Stability of sex differences in cognition in advanced old age: the role of education and attrition. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences, 61*(4), P245-P249.
 https://doi.org/10.1093/geronb/61.4P245
- Goldberg, T. E., Harvey, P. D., Wesnes, K. A., Snyder, P. J., & Schneider, L. S. (2015). Practice effects due to serial cognitive assessment: Implications for preclinical Alzheimer's disease randomized controlled trials. *Alzheimer's & Dementia*, 1(1), 103-111.

https://doi.org/10.1016/j.dadm.2014.11.003

- Hackett, R. A., Steptoe, A., Cadar, D., & Fancourt, D. (2019). Social engagement before and after dementia diagnosis in the English Longitudinal Study of Ageing. *PLoS One*, 14(8), e0220195. <u>https://doi.org/10.1371/journal.pone.0220195</u>
- Hahn, E. A., & Lachman, M. E. (2015). Everyday experiences of memory problems and control:
 The adaptive role of selective optimization with compensation in the context of memory
 decline. *Neuropsychology, Development, and Cognition. Section B, Aging, Neuropsychology and Cognition, 22*(1), 25-41.

https://doi.org/10.1080/13825585.2014.888391

- Heckhausen, J., Dixon, R. A., & Baltes, P. B. (1989). Gains and losses in development throughout adulthood as perceived by different adult age groups. *Developmental Psychology*, 25(1), 109-121. <u>https://doi.org/10.1037/0012-1649.25.1.109</u>
- Hendrie, H. C., Albert, M. S., Butters, M. A., Gao, S., Knopman, D. S., Launer, L. J., Yaffe, K., Cuthbert, B. N., Edwards, E., & Wagster, M. V. (2006). The NIH Cognitive and Emotional Health Project. Report of the critical evaluation study committee. *Alzheimer's & Dementia*, 2(1), 12-32. <u>https://doi.org/10.1016/j.jalz.2005.11.004</u>

Hertzog, C., Hülür, G., Gerstorf, D., & Pearman, A. (2018). Is subjective memory change in old age based on accurate monitoring of age-related memory change? Evidence from two longitudinal studies. *Psychology and Aging*, 33, 273-287.

https://doi.org/10.1037/pag0000232

- Hess, T. M. (2006). Attitudes toward aging and their effects on behavior. In UCLA Center on Aging (Ed.), *Handbook of the Psychology of Aging* (pp. 379-406). Elsevier.
- Hildesheim, C., Wahl, H.-W., Schröder, J., Kunzmann, U., Siebert, J. S., & Zenthöfer, A. (2019).
 Interdisciplinary Longitudinal Study on Adult Development and Aging. In D. Gu & M.
 E. Dupre (Eds.), *Encyclopedia of Gerontology and Population Aging*. Springer.
 https://doi.org/10.1007/978-3-319-69892-2_1098-1
- Hill, N. L., Mogle, J., Wion, R., Munoz, E., DePasquale, N., Yevchak, A. M., & Parisi, J. M. (2016). Subjective cognitive impairment and affective symptoms: A systematic review. *The Gerontologist*, 56(6), e109-e127. <u>https://doi.org/10.1093/geront/gn-w091</u>
- Hoyer, W. J., Stawski, R. S., Wasylyshyn, C., & Verhaeghen, P. (2004). Adult age and digit symbol substitution performance: A meta-analysis. *Psychology and Aging*, 19(1), 211. <u>https://doi.org/10.1037/0882-7974.19.1.211</u>
- Idler, E. L., & Benyamini, Y. (1997). Self-rated health and mortality: A review of twenty-seven community studies. *Journal of Health and Social Behavior*, 38(1), 21-37. https://doi.org/10.2307/2955359
- Jeste, D. V., Savla, G. N., Thompson, W. K., Vahia, I. V., Glorioso, D. K., Martin, A. S., Palmer,
 B. W., Rock, D., Golshan, S., Kraemer, H. C., & Depp, C. A. (2013, 2013/02/01).
 Association between older age and more successful aging: Critical role of resilience and

depression. American Journal of Psychiatry, 170(2), 188-196.

https://doi.org/10.1176/appi.ajp.2012.12030386

- Jokelainen, J., Timonen, M., Keinänen-Kiukaanniemi, S., Härkönen, P., Jurvelin, H., & Suija, K. (2019). Validation of the Zung self-rating depression scale (SDS) in older adults.
 Scandinavian Journal of Primary Health Care, 37(3), 353-357.
 https://doi.org/10.1080/02813432.2019.1639923
- Jonker, C., Geerlings, M. I., & Schmand, B. A. (2000). Are memory complaints predictive for dementia? A review of clinical and population-based studies. *International Journal of Geriatric Psychiatry*, 15(11), 983-991. <u>https://doi.org/10.1002/1099-1166(200011)15:11</u>
- Jylha, M., Guralnik, J. M., Balfour, J., & Fried, L. P. (2001). Walking difficulty, walking speed, and age as predictors of self-rated health: The women's health and aging study. *The Journals of Gerontology, Series A: Biological Sciences and Medical Sciences, 56*(10), M609-617. <u>https://doi.org/10.1093/gerona/56.10.M609</u>
- Kaspar, R., Gabrian, M., Brothers, A. F., Wahl, H.-W., & Diehl, M. K. (2019). Measuring awareness of age-related change: Development of a 10-Item short form for use in largescale surveys. *The Gerontologist*, 59(3), e130-e140. https://doi.org/10.1093/geront/gnx213
- Kennedy, N., Foy, K., Sherazi, R., McDonough, M., & McKeon, P. (2007). Long-term social functioning after depression treated by psychiatrists: A review. *Bipolar Disorders*, 9(1-2), 25-37. <u>https://doi.org/10.1111/j.1399-5618.2007.00326.x</u>
- Kessler, E.-M., Bowen, C. E., Baer, M., Froelich, L., & Wahl, H.-W. (2012). Dementia worry: A psychological examination of an unexplored phenomenon. *European Journal of Ageing*, 9(4), 275-284. <u>https://doi.org/10.1007/s10433-012-0242-8</u>

- Klaus, D., Engstler, H., Mahne, K., Wolff, J. K., Simonson, J., Wurm, S., & Tesch-Römer, C.
 (2017). Cohort Profile: The German Ageing Survey (DEAS). *International Journal of Epidemiology*, 46(4), 1105-1105g. <u>https://doi.org/10.1093/ije/dyw326</u>
- Klusmann, V., Sproesser, G., Wolff, J. K., & Renner, B. (2017). Positive self-perceptions of aging promote healthy eating behavior across the life span via social-cognitive processes. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences*, 1-10. <u>https://doi.org/10.1093/geronb/gbx139</u>
- Kornadt, A. E., Kessler, E.-M., Wurm, S., Bowen, C. E., Gabrian, M., & Klusmann, V. (2019). Views on ageing: A lifespan perspective. *European Journal of Ageing*. https://doi.org/10.1007/s10433-019-00535-9
- Kotter-Grühn, D., & Hess, T. M. (2012). The impact of age stereotypes on self-perceptions of aging across the adult lifespan. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences*, 67, 563-571. <u>https://doi.org/10.1093/geronb/gbr153</u>
- Lawton, M. P. (1975). The Philadelphia Geriatric Center Morale Scale: A revision. *Journal of Gerontology*, *30*(1), 85-89.
- Levy, B. R. (1994). Aging-Associated Cognitive Decline. *International Psychogeriatrics*, 6(1), 63-68. <u>https://doi.org/10.1017/S1041610294001626</u>
- Levy, B. R., Pilver, C., Chung, P. H., & Slade, M. D. (2014). Subliminal strengthening: Improving older individuals' physical function over time with an implicit-age-stereotype intervention. *Psychological Science*, 25(12), 2127-2135.

https://doi.org/10.1177/0956797614551970

- Levy, B. R., Slade, M. D., Pietrzak, R. H., & Ferrucci, L. (2018). Positive age beliefs protect against dementia even among elders with high-risk gene. *PLOS ONE*, *13*(2), e0191004. <u>https://doi.org/10.1371/journal.pone.0191004</u>
- Martyr, A., & Clare, L. (2012). Executive function and activities of daily living in Alzheimer's disease: A correlational meta-analysis. *Dementia and Geriatric Cognitive Disorders*, 33(2-3), 189-203. <u>https://doi.org/10.1159/000338233</u>
- Menkin, J. A., Smith, J. L., & Bihary, J. G. (2020). Brief anti-ageism messaging effects on physical activity motivation among older adults. *Journal of Applied Gerontology*, 0733464820960925. <u>https://doi.org/10.1177/0733464820960925</u>
- Miche, M., Elsässer, V. C., Schilling, O. K., & Wahl, H.-W. (2014a). Attitude toward own aging in midlife and early old age over a 12-year period: Examination of measurement equivalence and developmental trajectories. *Psychology and Aging, 29*(3), 588-600.
 https://doi.org/10.1037/a0037259
- Miche, M., Wahl, H.-W., Diehl, M. K., Oswald, F., Kaspar, R., & Kolb, M. (2014b). Natural occurrence of subjective aging experiences in community-dwelling older adults. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences, 69*(2), 174-187. https://doi.org/10.1093/geronb/gbs164
- Mol, M. E., Ruiter, R. A., Verhey, F. R., Dijkstra, J., & Jolles, J. (2008). A study into the psychosocial determinants of perceived forgetfulness: Implications for future interventions. *Aging & Mental Health*, 12(2), 167-176.

https://doi.org/10.1080/13607860801972503

- Montepare, J. M. (2019). An exploration of subjective age, actual age, age awareness, and engagement in everyday behaviors. *European Journal of Ageing*, 299-307. <u>https://doi.org/10.1007/s10433-019-00534-w</u>
- Muthén, L. K., & Muthén, B. O. (2017). Mplus User's Guide.
- O'Brien, E. L., Torres, G. E., & Neupert, S. D. (2020, Apr 23). Cognitive interference in the context of daily stressors, daily awareness of age-related change, and general aging attitudes. *The Journals of Gerontology, Psychological Sciences: Series B*, 76(5), 920-929. <u>https://doi.org/10.1093/geronb/gbaa155</u>
- Oswald, W., & Fleischmann, U. (1995). Nürnberger-Alters-Inventar (NAI). 3. überarbeitete und ergänzte Auflage. Hogrefe.
- Palsson, S. P., Ostling, S., & Skoog, I. (2001). The incidence of first-onset depression in a population followed from the age of 70 to 85. *Psychological Medicine*, 31(7), 1159-1168. https://doi.org/10.1017/S0033291701004524
- Park, D. C., & Reuter-Lorenz, P. (2009). The adaptive brain: Aging and neurocognitive scaffolding. *Annual Review of Psychology*, 60, 173-196. https://doi.org/10.1146/annurev.psych.59.103006.093656
- Preacher, K. J., Wichman, A. L., MacCallum, R. C., & Briggs, N. E. (2008). *Latent Growth Curve Modeling* (V. Knight, Ed.). Sage Publications.

https://doi.org/10.4135/9781412984737

Puth, M.-T., Weckbecker, K., Schmid, M., & Münster, E. (2017). Prevalence of multimorbidity in Germany: impact of age and educational level in a cross-sectional study on 19,294 adults. *BMC public health*, 17(1), 826-826. <u>https://doi.org/10.1186/s12889-017-4833-3</u>

- Rakesh, G., Szabo, S. T., Alexopoulos, G. S., & Zannas, A. S. (2017). Strategies for dementia prevention: Latest evidence and implications. *Therapeutic Advances in Chronic Disease*, 8(8-9), 121-136. <u>https://doi.org/10.1177/2040622317712442</u>
- Robertson, D. A., & Kenny, R. A. (2016). Negative perceptions of aging modify the association between frailty and cognitive function in older adults. *Personality and Individual Differences*, 100, 120-125. <u>https://doi.org/10.1016/j.paid.2015.12.010</u>
- Royall, D. R., Lauterbach, E. C., Kaufer, D., Malloy, P., Coburn, K. L., & Black, K. J. (2007). The cognitive correlates of functional status: A Review from the committee on research of the American Neuropsychiatric Association. *The Journal of Neuropsychiatry and Clinical Neurosciences*, 19(3), 249-265. <u>https://doi.org/10.1176/jnp.2007.19.3.249</u>
- Sabatini, S., Obioha, C. U., Ballard, C., Collins, R., Corbett, A., Brooker, H., & Clare, L. (2021a). Exploring Awareness of Age-Related Changes among over 50s in the UK: Findings from the PROTECT study. *In Press*.
- Sabatini, S., Silarova, B., Martyr, A., Collins, R., Ballard, C., Anstey, K. J., Kim, S., & Clare, L. (2020a). Associations of awareness of age-related change with emotional and physical well-being: A systematic review and meta-analysis. *The Gerontologist*, 60(6), e477-e490. <u>https://doi.org/10.1093/geront/gnz101</u>
- Sabatini, S., Ukoumunne, O. C., Ballard, C., Brothers, A. F., Kaspar, R., Collins, R., Kim, S., Corbett, A., Aarsland, D., Hampshire, A., Brooker, H., & Clare, L. (2020b). International relevance of two measures of awareness of age-related change (AARC). *BMC Geriatrics,* 20(1), 359. https://doi.org/10.1186/s12877-020-01767-6
- Sabatini, S., Ukoumunne, O. C., Ballard, C., Collins, R., Anstey, K. J., Diehl, M. K., Brothers, A. F., Wahl, H.-W., Corbett, A., Hampshire, A., Brooker, H., & Clare, L. (2021b). Cross-

sectional association between objective cognitive performance and perceived age-related gains and losses in cognition. *International Psychogeriatrics*, *33*, 727-741. https://doi.org/10.1017/S1041610221000375

- Sabatini, S., Ukoumunne, O. C., Ballard, C., Collins, R., Corbett, A., Brooker, H., & Clare, L. Cross-sectional and longitudinal associations between subjective sleep difficulties and self-perceptions of aging. *Behavioral Sleep Medicine*.
- Sabatini, S., Ukoumunne, O. C., Ballard, C., Collins, R., Corbett, A., Brooker, H., & Clare, L. (2020c, Aug). The cross-sectional relationship between pain and awareness of age-related changes. *British Journal of Pain*, 15(3), 335-344.

https://doi.org/10.1177/2049463720961798

Sabatini, S., Ukoumunne, O. C., Ballard, C., Collins, R., Kim, S., Corbett, A., Aarsland, D., Hampshire, A., Brooker, H., & Clare, L. (2021c). What does feeling younger or older than one's chronological age mean to men and women? Qualitative and quantitative findings from the PROTECT study. *Psychology & Health*.

https://doi.org/10.1080/08870446.2021.1960989

- Sabatini, S., Ukoumunne, O. C., Ballard, C., Diehl, M. K., Wahl, H.-W., Brothers, A. F., & Clare, L. (2020d). Health Differences Among People With Varying Profiles of Awareness of Positive and Negative Age-Related Changes. *Innovation in Aging,* 4(Supplement_1), 480-480.
- Sabatini, S., Ukoumunne, O. C., Martyr, A., Ballard, C., Lamont, R. A., Collins, R., Pentecost, C., Hunt, A., Quinn, C., Thom, J. M., & Clare, L. (2021d). Relationship between selfperceptions of aging and 'living well' among people with mild-to-moderate dementia:

Findings from the ideal programme. *Archives of Gerontology and Geriatrics*, *94*, 104328. https://doi.org/10.1016/j.archger.2020.104328

- Sabatini, S., Woods, R. T., Ukoumunne, O. C., Ballard, C., Collins, R., & Clare, L. (2021e).
 Associations of subjective cognitive and memory decline with depression, anxiety, and two-year change in objectively-assessed global cognition and memory. *Aging, Neuropsychology, and Cognition*, 1-27. <u>https://doi.org/10.1080/13825585.2021.1923634</u>
- Sargent-Cox, K. A., Anstey, K. J., & Luszcz, M. A. (2012). Change in health and selfperceptions of aging over 16 years: The role of psychological resources. *Health Psychology*, 31(4), 423-432. https://doi.org/10.1037/a0027464
- Sarkisian, C. A., Prohaska, T. R., Davis, C., & Weiner, B. (2007). Pilot test of an attribution retraining intervention to raise walking levels in sedentary older adults. *Journal of the American Geriatrics Society*, 55, 1842-1846. <u>https://doi.org/10.1111/j.1532-</u> 5415.2007.01427.x
- Schönstein, A., Dallmeier, D., Denkinger, M., Rothenbacher, D., Klenk, J., Bahrmann, A., &
 Wahl, H.-W. (2021). Health and Subjective Views on Aging: Longitudinal Findings from the ActiFE Ulm Study. *The Journals of Gerontology: Series B*.
 https://doi.org/10.1093/geronb/gbab023
- Seidler, A. L., & Wolff, J. K. (2017). Bidirectional associations between self-perceptions of aging and processing speed across 3 years. *The Journal of Gerontopsychology and Geriatric Psychiatry*, 30, 49-59. <u>https://doi.org/10.1024/1662-9647/a000165</u>
- Selig, J. P., & Little, T. D. (2012). Autoregressive and cross-lagged panel analysis for longitudinal data. In B. Laursen, T. D. Little, & N. A. Card (Eds.), *Handbook of developmental research methods* (pp. <u>265–278</u>). The Guilford Press.

- Siebert, J. S., Braun, T., & Wahl, H. W. (2020). Change in attitudes toward aging: Cognitive complaints matter more than objective performance. *Psychology and Aging*, 35(3), 357-368. <u>https://doi.org/10.1037/pag0000451</u>
- Siebert, J. S., Wahl, H.-W., & Schröder, J. (2016). The role of attitude toward own aging for fluid and crystallized functioning: 12-year evidence from the ILSE study. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences, 73*(5), 836-845.
 https://doi.org/10.1093/geronb/gbw050
- Spuling, S. M., Miche, M., Wurm, S., & Wahl, H.-W. (2013). Exploring the causal interplay of subjective age and health dimensions in the second half of life: A cross-lagged panel analysis. *Zeitschrift für Gesundheitspsychologie*, 21(1), 5-15. https://doi.org/10.1026/0943-8149/a000084
- Staats, S., Heaphey, K., Miller, D., Partlo, C., Romine, N., & Stubbs, K. (1993). Subjective age and health perceptions of older persons: Maintaining the youthful bias in sickness and in health. *The International Journal of Aging and Human Development*, 37(3), 191-203. https://doi.org/10.2190/373B-PJ6U-DWAA-4K03

StataCorp. (2017). Stata Statistical Software: Release 16. College Station, TX2016.

- Stephan, Y., Sutin, A. R., & Terracciano, A. (2015). How old do you feel? The role of age discrimination and biological aging in subjective age. *PloS one*, 10(3), e0119293. <u>https://doi.org/10.1371/journal.pone.0119293</u>
- Steverink, N., Westerhof, G. J., Bode, C., & Dittmann-Kohli, F. (2001). The personal experience of aging, individual resources, and subjective well-being. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences, 56*(6), 364-373. https://doi.org/10.1093/geronb/56.6.P364

- Stuck, A. E., Walthert, J. M., Nikolaus, T., Büla, C. J., Hohmann, C., & Beck, J. C. (1999). Risk factors for functional status decline in community-living elderly people: A systematic literature review. *Social Science & Medicine*, 48(4), 445-469. https://doi.org/10.1016/S0277-9636(98)00370-0
- Sun, J. K., & Smith, J. (2017). Self-perceptions of aging and perceived barriers to care: Reasons for health care delay. *The Gerontologist*, 57(suppl_2), S216-S226. https://doi.org/10.1093/geront/gnx014
- Tewes, U. (1991). *HAWIE-R. Hamburg-Wechsler Intelligenztest für Erwachsene. Revision 1991*. Huber.
- Vailati Riboni, F., Comazzi, B., Castelnuovo, G., Molinari, E., & Pagnini, F. (2018). Mindful age and technology: Promoting quality of life in older adults with a tablet/smartphone app. Pervasive Computing Paradigms for Mental Health, Cham.
- Vailati Riboni, F., Sadowski, I., Comazzi, B., & Pagnini, F. (2020). Mindful age and technology: A qualitative analysis of a tablet/smartphone app intervention designed for older adults. *Integrative Psychological and Behavioral Science*. <u>https://doi.org/10.1007/s12124-020-</u>09580-x
- Wahl, H.-W., & Ehni, H.-J. (2020). Advanced old age as a developmental dilemma: An in-depth comparison of established fourth age conceptualizations. *Journal of Aging Studies*, 55, 100896. <u>https://doi.org/10.1016/j.jaging.2020.100896</u>
- Ware, J. E. J., & Sherbourne, C. D. (1992). The MOS 36-item short-form health survey (SF-36). Conceptual framework and item selection. *Medical Care*, 30(6), 473-483. <u>https://doi.org/10.1177/016327879702000102</u>

- Wechsler, D. (1981). WAIS-R : Wechsler adult intelligence scale-revised. Psychological Corporation. <u>https://search.library.wisc.edu/catalog/999605091402121</u>
- Wermelinger Ávila, M. P., Lucchetti, A. L. G., & Lucchetti, G. (2017). Association between depression and resilience in older adults: A systematic review and meta-analysis.
 International Journal of Geriatric Psychiatry, 32(3), 237-246.
 https://doi.org/10.1002/gps.4619
- Westerhof, G. J., Barrett, A. E., & Steverink, N. (2003). Forever young?: A comparison of age identities in the United States and Germany. *Research on Aging*, 25(4), 366-383. <u>https://doi.org/10.1177/0164027503025004002</u>
- Westerhof, G. J., Gabrian, M., Brothers, A. F., Barrett, A. E., Diehl, M. K., Montepare, J. M., Wahl, H.-W., & Wurm, S. (2014). The influence of subjective aging on health and longevity: A meta-analysis of longitudinal data. *Psychology and Aging, 29*, 793-802. https://doi.org/10.1037/a0038016
- Wettstein, M., Schilling, O. K., & Wahl, H.-W. (2016). "Still feeling healthy after all these years": The paradox of subjective stability versus objective decline in very old adults' health and functioning across five years. *Psychology and Aging*, *31*(8), 815. https://doi.org/10.1037/pag0000137
- Weyerer, S., Eifflaender-Gorfer, S., Wiese, B., Luppa, M., Pentzek, M., Bickel, H., Bachmann, C., Scherer, M., Maier, W., & Riedel-Heller, S. G. (2013). Incidence and predictors of depression in non-demented primary care attenders aged 75 years and older: Results from a 3-year follow-up study. *Age and Ageing*, *42*(2), 173-180.

https://doi.org/10.1093/ageing/afs184

Wolff, J. K., Warner, L. M., Ziegelmann, J. P., & Wurm, S. (2014). What do targeting positive views on ageing add to a physical activity intervention in older adults? Results from a randomised controlled trial. *Psychology & Health, 29*(8), 915-932.

https://doi.org/10.1080/08870446.2014.896464

- World Health Organization. (2018). *Gender and Health Survey*. <u>https://www.euro.who.int/__data/assets/pdf_file/0007/382408/gender-health-report-</u> eng.pdf
- Zanini, S., Voltolini, A., Gragnano, G., Fumagalli, E., & Pagnini, F. (2018). Changes in pain perception following psychotherapy: The mediating role of psychological components.
 Pain Research & Management, 2018, 8713084-8713084.
 https://doi.org/10.1155/2018/8713084
- Zhang, S., & Neupert, S. D. (2020). Within-and between-person relationships among health, awareness of aging, and control beliefs: A micro-longitudinal study. *The Journals of Gerontology, Series B: Psychological Sciences and Social Sciences*. https://doi.org/10.1093/geronb/gbaa180

Zhu, X., & Neupert, S. D. (2020, Jun). Dynamic awareness of age-related losses predict concurrent and subsequent changes in daily inductive reasoning performance. *British Journal of Developmental Psychology*, 39(2), 282-298.

https://doi.org/10.1111/bjdp.12344

- Zimprich, D., & Kurtz, T. (2015). Subjective and objective memory changes in old age across five years. *Gerontology*, 61(3), 223-231. <u>https://doi.org/10.1159/000369927</u>
- Zung, W. W. K. (1965). A self-rating depression scale. Archives of General Psychiatry, 12(1), 63-70. https://doi.org/10.1001/archpsyc.1965.01720310065008

Characteristics	Study sample (N = 103)	Missing values
	N (%)	Ν
Women	51 (49.5%)	0
Education	13.53 (2.59; 9 to 18)	1
Up to 10 years of education	10 (9.8%)	
Between 11 and 12 years	33 (32.4%)	
Between 13 and 15 years	36 (35.3%)	
More than 15 years of school and university education	23 (22.5%)	
	Mean (SD; range)	
Age at baseline	62.77 (0.97; 61 to 64)	0
AARC-50 gains at follow-up	78.31 (13.79; 33 to 115)	0
AARC-50 losses at follow-up	61.02 (14.37; 28 to 94)	0
Attitude toward own aging		
Baseline	1.78 (0.28; 1 to 2)	2
20-year follow-up	1.57 (0.31; 1 to 2)	1
Objectively assessed physical health		
Baseline	3.83 (0.70; 2 to 5)	4
4-year follow-up	3.68 (0.51; 2 to 4)	0
12-year follow-up	3.76 (0.67; 2 to 5)	3
20-year follow-up	3.15 (0.89; 1 to 4)	57
Subjective health rating		
Baseline	3.96 (0.81; 1 to 5)	2
4-year follow-up	3.87 (0.85; 1 to 5)	0
12-year follow-up	3.75 (0.90; 1 to 5)	4
20-year follow-up	3.30 (1.02; 1 to 5)	2
Digit span		
Baseline	11.11 (1.65; 7 to 16)	0
4-year follow-up	10.92 (1.73; 8 to 16)	0
12-year follow-up	10.76 (1.95; 6 to 15)	4
20-year follow-up	9.88 (1.68; 6 to 14)	5
Digit symbol		
Baseline	47.69 (10.08; 27 to 67)	0
4-year follow-up	47.46 (10.43; 25 to 67)	0
12-year follow-up	43.54 (9.98; 24 to 67)	4
20-year follow-up	35.32 (9.97; 20 to 65)	8

Table 1. Descriptive statistics of study variables

Information		
Baseline	17.29 (3.69; 8 to 24)	0
4-year follow-up	17.48 (3.81; 8 to 23)	0
12-year follow-up	17.32 (3.65; 8 to 23)	4
20-year follow-up	22.671 (7.94, 0 to 44)	9
Similarities		
Baseline	26.63 (3.70; 16 to 32)	0
4-year follow-up	25.83 (4.86; 9 to 31)	0
12-year follow-up	25.96 (4.96; 7 to 32)	4
20-year follow-up	16.18 (4.23; 5 to 23)	9
Block design		
Baseline	29.20 (7.73; 13 to 47)	0
4-year follow-up	26.77 (8.27; 6 to 45)	0
12-year follow-up	26.91 (7.17; 11 to 40)	4
20-year follow-up	25.11 (4.58; 6 to 31)	10
Picture completion		
Baseline	12.77 (3.15; 1 to 17)	0
4-year follow-up	13.05 (2.91; 2 to 17)	0
12-year follow-up	12.99 (3.15; 1 to 17)	4
20-year follow-up	12.1 (3.50; 0 to 17)	12
Depressive symptoms		
Baseline	1.65 (0.32; 1.05 to 2.45)	3
4-year follow-up	1.63 (0.30; 1 to 2.4)	0
12-year follow-up	1.70 (0.33; 1.1 to 2.6)	5
20-year follow-up	1.87 (0.34; 1.25 to 3.0)	2

Health indicators	Mean change from baseline to 20-year follow-up (SD); 95% CI	t-test (df); <i>p</i> -value
Change in digit span	1.19 (1.60); 0.87 to 1.51	7.40 (97); < .001
Change in digit symbol	12.48 (8.16); 10.82 to 14.15	14.90 (94); < .001
Change in information	-5.32 (7.66); -6.89 to -3.75	-6.73 (93); < .001
Change in similarities	10.44 (4.21); 9.57 to 11.30	24.04 (93); < .001
Change in block design	4.09 (8.59); 2.32 to 5.85	4.59 (92); < .001
Change in picture completion	0.85 (3.08); 0.20 to 1.49	2.62 (90); .005
Change in objectively assessed physical health	0.61 (1.02); 0.30 to 0.92	4.00 (43); < .001
Change in subjective health rating	0.64 (1.02); 0.43 to 0.84	6.18 (98); < .001
Change in depression	-0.22 (0.36); -0.29 to -0.14	-5.93 (97); < .001

Table 2. 20-year change in indicators of cognitive functioning, physical health, and mental health

Table 3. Bivariate correlations among AARC-gains, AARC-losses, and ATOA at 20-year follow-up with baseline and 20-year

follow-up scores on indicators of cognitive functioning, physical health, and mental health

Indicators of cognitive functioning, physical and mental health	AARC-gains	AARC-losses	АТОА
	<i>r; p</i> -value	<i>r; p</i> -value	<i>r; p</i> -value
Baseline digit span	.05; .61	26; .01	.25; .01
Follow-up digit span	.003; .98	29; .003	.25; .01
Baseline digit symbol	03; .79	.002; .98	.002; .98
Follow-up digit symbol	.10; .34	23; .02	.16; .13
Baseline information	.03; .79	.14; .17	.14; .17
Follow-up information	.12; .24	12; .24	.19; .07
Baseline similarities	.15; .12	.08; .44	.08; .44
Follow-up similarities	.12; .23	20; .05	.15; .15
Baseline block design	.02; .84	.03; .77	.03; .77
Follow-up block design	.18; .08	17; .10	.07; .47
Baseline picture completion	.07; .49	.18; .07	.18; .07
Follow-up picture completion	.14; .20	10; .36	.20; .05
Baseline objectively assessed physical health	.01; .89	25; .01	.39; .003
Follow-up objectively assessed physical health	12; .41	39; .04	.45; .002
Baseline subjective health rating	12; .23	25; .01	.60; .001
Follow-up subjective health rating	31; .002	61; .001	.28; .004
Baseline depression	.08; .46	.33; .001	34; .001
Follow-up depression	.05; .62	.55; .001	76; .001

	AARC-gains		AARC-losses		ATOA	
Predictor	β (95% CI); <i>p</i>	R^2	β (95% CI); <i>p</i>	R^2	β (95% CI); <i>p</i>	R^2
20-year change in digit span	.08 (12 to .27); .451	1%	.01 (19 to .21); .907	0.01%	002 (20 to .20); .983	0%
20-year change in digit symbol	26 (45 to08); .005	7%	.18 (01 to .37); .070	3%	14 (34 to .05); .150	2%
20-year change in information	09 (29 to .11); .384	1%	.07 (13 to .27); .514	1%	13 (33 to .06); .188	2%
20-year change in similarities	.08 (12 to .28); .443	1%	.14 (05 to .34); .150	2%	11 (31 to .09); .287	0.1%
20-year change in block design	11 (31 to .09); .282	1%	.06 (14 to .26); .573	0.3%	02 (22 to .18); .838	0.002%
20-year change in picture completion	16 (36 to .04); .115	3%	20 (40 to01); .042	4%	07 (27 to .14); .532	0.03%
20-year change in objectively assessed	.08 (21 to .37); .597	1%	01 (30 to .29); .959	0.01%	20 (48 to .08); .162	0.04%
physical health						
20-year change in subjective health rating	.09 (10 to .29); .364	1%	.35 (.18 to .52); <.001	12%	40 (56 to24); <.001	0.15%
20-year change in depressive symptoms	.11 (09 to .30); .286	1%	28 (46 to10); <.001	8%	.43 (.28 to .59); <.001	19%

 Table 4. Univariate linear regressions with 20-year change in indicators of cognitive functioning, physical health, and mental health as predictors of AARC-gains, AARC-losses, or ATOA: Standardized regression coefficients

Notes: β = Standardized regression coefficient. Partial R² = Partial R-squared/coefficient of determination. Significant *p*-value after applying Bonferroni's correction = .002.

	Predictors	β (95% CI); <i>p</i>	$R^2/Partial R^2$
Simple regression	20-year change in digit symbol	26 (45 to08); .005	7%
Adjusted for demographic variables	20-year change in digit symbol	24 (43 to05); .012	6%
	Sex	.06 (14 to .25); .551	0.3%
	Education	11 (31 to .08); .258	1%
Adjusted for demographic variables, perceived health, and	20-year change in digit symbol	24 (43 to06); .011	6%
depression	Sex	02 (19 to .22); .871	0.03%
	Education	15 (35 to .05); .141	2%
	Subjective health rating	18 (38 to .02); .075	3%
	Depression	01 (22 to .20); .924	0.01%
20-year change in s	subjective health rating as predictor of AARC-	losses	
Simple regression	20-year change in subjective health rating	.35 (.18 to .52); <.001	12%
Adjusted for demographic variables	20-year change in subjective health rating	.36 (.20 to .53); <.001	13%
	Sex	.05 (13 to .24); .570	0.3%
	Education	10 (28 to .09); .302	1%
Adjusted for demographic variables, subjective health rating,	20-year change in subjective health rating	.38 (.22 to .53); <.001	14%
depression	Sex	05 (22 to .13); .607	0.2%
	Education	05 (22 to .12); .573	0.2%
	Depression	.38 (.21 to .54); <.001	13%
20-year chang	ge in digit symbol as predictor of AARC-losses	5	
Simple regression	20-year change in digit symbol	.18 (01 to .37); .070	3%
Adjusted for demographic variables	20-year change in digit symbol	.20 (.01 to .39); .044	4%

Table 5. Multiple regressions with 20-year change in indicators of cognitive functioning, physical health, and mental health as predictors of AARC-gains, AARC-losses, or ATOA: Standardized regression coefficients

	Sex	.09 (11 to .29); .383	1%
	Education	12 (32 to .08); .229	1%
Adjusted for demographic variables, subjective health rating,	20-year change in digit symbol	.20 (.02 to .38); .027	4%
depression	Sex	04 (23 to .15); .697	0.1%
	Education	07 (26 to .12); .473	0.4%
	Perceived health	18 (37 to .001); .051	3%
	Depression	.35 (.17 to .53); <.001	01%
20-year change i	n picture completion as predictor of AARC-lo	osses	
Simple regression	20-year change in picture completion	20 (40 to01); .042	4%
Adjusted for demographic variables	20-year change in picture completion	19 (39 to .01); .064	3%
	Sex	.06 (14 to .26); .563	0.4%
	Education	08 (29 to .12); .429	1%
Adjusted for demographic variables, subjective health rating,	20-year change in picture completion	11 (31 to .08); .257	1%
depression	Sex	05 (24 to .15); .635	0.2%
	Education	04 (23 to .16); .707	0.1%
	Perceived health	17 (36 to .03); .100	2%
	Depression	.36 (.17 to .54); <.001	11%
20-year change in	depressive symptoms as predictor of AARC-	losses	
Simple regression	20-year change in depressive symptoms	28 (46 to10); <.001	8%
Adjusted for demographic variables	20-year change in depressive symptoms	30 (48 to12); .001	9%
	Sex	.10 (09 to .29); .311	1%
	Education	11 (30 to .08); .275	1%
Adjusted for demographic variables, subjective health rating,	20-year change in depressive symptoms	25 (43 to07); .006	6%
depression	Sex	.05 (14 to .24); .578	0.03%
	Education	10 (29 to .08); .277	1%
	Subjective health rating	25 (43 to07); .007	6%

c	in perceived health as predictor of ATOA		
		β (95% CI); <i>p</i>	$R^2/Partial R^2$
Simple regression	Change in subjective health rating	40 (56 to24); <.001	0.15%
Adjusted for demographic variables	Change in subjective health rating	45 (60 to30); <.001	20%
	Sex	07 (25 to .11); .462	0.4%
	Education	003 (18 to .18); .978	0%
Adjusted for demographic variables, subjective health rating,	Change in subjective health rating	45 (59 to31); <.001	20%
and depression	Sex	.01 (16 to .19); .875	0.02%
	Education	04 (21 to .13); .648	0.02%
	Depression	32 (48 to15); <.001	9%
Change in	depressive symptoms as predictor of ATOA		
Simple regression	Change in depressive symptoms	.43 (.28 to .59); <.001	19%
Adjusted for demographic variables	Change in depressive symptoms	.46 (.31 to .61); <.001	21%
	Sex	15 (32 to .03); .107	2%
	Education	.02 (16 to .20); .841	0.03%
Adjusted for demographic variables, subjective health rating,	Change in depressive symptoms	.44 (.38 to .59); <.001	18%
and depression	Sex	12 (31 to .06); .178	1%
	Education	.02 (16 to .19); .856	0.03%
	Subjective health rating	.12 (06 to .30); .171	1%

Notes: β = Standardized regression coefficient. Partial R² = Partial R-squared/coefficient of determination.